Author’s response to reviews

Title: Outcome of infections due to pandrug-resistant (PDR) Gram-negative bacteria

Authors:

Matthew E. Falagas (matthew.falagas@tufts.edu)
Ioannis A. Bliziotis (i.bliziotis@alfahc.gr)
Sofia K. Kasiakou (s.kasiakou@alfahc.gr)
George Samonis (alfa.healthcare@alfahc.gr)
Panayiota Athanassopoulou (alfa.healthcare@alfahc.gr)
Argyris Michalopoulos (amichalopoulos@hol.gr)

Version: 2  Date: 17 February 2005

Author’s response to reviews: see over
Athens, February 17, 2005

To: Editor, *BMC Infectious Diseases*

Title: Outcome of infections due to pandrug-resistant (PDR) Gram-negative bacteria

Authors: Matthew E. Falagas, M.D., M.Sc., Ioannis A. Bliziotis, M.D., Sofia K. Kasiakou, M.D., George Samonis, M.D., Panayiota Athanassopoulou M.D., Argyris Michalopoulos, M.D.

Dear Editor,

The abstract was modified based on your suggestions. Thank you for considering our paper for publication in *BMC Infectious Diseases*.

Sincerely,

Matthew E. Falagas, M.D. M.Sc.
Adjunct Assistant Professor of Medicine, Tufts University School of Medicine, Boston, Massachusetts
President, Board of Trustees, Alfa Institute of Biomedical Sciences, Athens, Greece
Director, Infectious Diseases Clinic, "Henry Dunant” Hospital, Athens, Greece